Author’s response to reviews

Title: ELECTRONIC HEALTH RECORDS ACCURATELY PREDICT RENAL REPLACEMENT THERAPY IN ACUTE KIDNEY INJURY

Authors:

Sanmay Low (sanmay_low@nuhs.edu.sg)
Anantharaman Vathsala (vathsala_anantharaman@nuhs.edu.sg)
Tanushya Murali (mdctmm@nus.edu.sg)
Long Pang (pang_long@nuhs.edu.sg)
Graeme MacLaren (graeme_maclaren@nuhs.edu.sg)
Wan Ying Ng (a0116891@u.nus.edu)
Sabrina Haroon (sabrina_haroon@nuhs.edu.sg)
Amartya Mukhopadhyay (amartya_mukhopadhyay@nuhs.edu.sg)
Shir Lynn Lim (shir_lynn_lim@nuhs.edu.sg)
Bee-Hong Tan (bee_hong_tan@nuhs.edu.sg)
Titus Lau (titus_lau@nuhs.edu.sg)
Horng-Ruey Chua (horng_ruey_chua@nuhs.edu.sg)

Version: 1 Date: 02 Jan 2019

Author’s response to reviews:

Dear Editors and Reviewers,

Thank you for taking the time to review our manuscript. We have made the necessary amendments and sincerely hope this version could be reviewed and considered favorably for publication in your journal.

We have attached below a point-by-point response as requested, with the editor and reviewers’ comments and our response below.
Editor’s comments:

1. Please include a statement in the Authors’ contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case.
   • We have edited this section to include that all authors have read and approved the manuscript.

2. In the section 'Funding', please also describe the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.
   • The funding body did not have any role in the design, analysis, interpretation of the data and in the writing of the manuscript, and this has been included in the manuscript.

Reviewer’s comments:

1. In both the title and first sentence of the abstracts, "Electronic health records accurately predicts..." should be "Electronic health records accurately predict..."

2. In at least one place, autocorrect has changed EHR to HER.

3. Change "biasness" to "bias" in the limitations.
   • We have edited the grammar in the manuscript according to the reviewer’s comments.